Author's response to reviews

Title: Accuracy of self-reported height measurements in parents and its effect on mid-parental target height calculation

Authors:

Ieva Braziuniene (Braziuniene-leva@cooperhealth.edu)
Thomas A Wilson (tawilson@mail.som.sunysb.edu)
Andrew H Lane (andrew.lane@stonybrook.edu)

Version: 5 Date: 2 April 2007

Author's response to reviews: see over
Dear Sirs,

Thank you for accepting in principle our manuscript # 1893570862125025, “Accuracy of self-reported height measurements in parents and its effect on mid-parental target height calculation” for BioMed Central. I have just resubmitted the manuscript with the requested final figure cropping and added corresponding author information.

After many hours at the computer, I cannot figure out why the EndNote program would only extract 14 of the 17 references. I tried remaking the library and re-entering the references – I am not sure if you will now be able to extract all 17 references. Even if you are still not able to, the manuscript as currently submitted has all 17 of the references listed properly in the reference section. Therefore, although they may or may not be extractable, my manuscript should be publishable as is.

Sincerely,

Andrew H. Lane, M.D.